Author’s response to reviews

**Title:** Obesity and elevated blood pressure among adolescents in Lagos, Nigeria: A cross-sectional study.

**Authors:**

- Abiola O Oduwole (bioodu@yahoo.com)
- Taiwo A Ladapo (drteeladapo@yahoo.com)
- Iretiola B Fajolu (iretifajolu@justice.com)
- Ekanem N Ekure (ekaekure@yahoo.com)
- Olufunmilayo Funke Adeniyi (layo_funke@yahoo.co.uk)

**Version:** 4  **Date:** 6 July 2012

**Author’s response to reviews:**

Dear Editor,

I hereby apologise for the password protection on the article.

This has been removed.

Regards,

Dr Ladapo